

**Author's response to reviews**

**Title:** Network Analysis of Global Tobacco Control Collaboration: Data from the World Conference on Tobacco or Health (WCTOH)

**Authors:**

Scott Leischow (leischow.scott@mayo.edu)

Janet Okamoto (okamoto.janet@mayo.edu)

Scott McIntosh (scott\_mcintosh@urmc.rochester.edu)

Deborah Ossip (deborah\_ossip@urmc.rochester.edu)

Harry Lando (lando001@umn.edu)

**Version:** 1 **Date:** 09 Nov 2016

**Author's response to reviews:**

We revised the paper according the criterion indicated by the editors.